Author's response to reviews

Title: Cost and efficiency of public sector sexually transmitted infection clinics in Andhra Pradesh, India

Authors:

- Lalit Dandona (dandona@asci.org.in)
- Pratap Sisodia (pratapsisodia@asci.org.in)
- T.I.n. Prasad (drellen2002@yahoo.com)
- Elliot Marseille (emarseille@comcast.net)
- M. Chalapathi Rao (mcrao@asci.org.in)
- A. Anod Kumar (anod@asci.org.in)
- S.G. Prem Kumar (prem@asci.org.in)
- Y.k. Ramesh (ramesh@asci.org.in)
- Mead Over (meadover@worldbank.org)
- M. Someshwar (somesh@asci.org.in)
- James G. Kahn (jgkahn@itsa.ucsf.edu)

Version: 3 Date: 28 October 2005

Author's response to reviews:

28 October 2005

Editors
BMC Health Services Research

MS 4276605196681967 re-revision

Dear Editors,

The minor comment by Reijo Sund on our revised manuscript is addressed as follows in today’s further revision of this manuscript:

Page 12 bottom. 14122 is standard error of the estimate (not of R square). To avoid confusion, we have removed this, as suggested by the reviewer. Also, R square value of "1.00" is now replaced with "almost 1.00", as suggested by the reviewer.

We now look forward to a rapid final editorial decision on our manuscript, and thank the reviewers and editors for inputs on this manuscript.

Kind regards,
Lalit Dandona

Lalit Dandona, MD, MPH
Professor
Chairperson, Health Studies Area
Director, Centre for Human Development
Administrative Staff College of India
Raj Bhavan Road, Hyderabad - 500 082, India
Email: dandona@asci.org.in, lalit.dandona@rediffmail.com
Tel: (+91 40) 2337 6958, 2331 0952
Fax: (+91 40) 2331 2954